Author’s response to reviews

Title: Dietary diversity and associated factors among children aged 6-23months in Sinan Woreda, Northwest Ethiopia,2016:cross-sectional study

Authors:

Habtamu temesgen (habtamutem@gmail.com)
te bikew yeneabat (wait4myfather@gmail.com)
Muluken Teshome (mulukenteshome@gmail.com)

Version: 4 Date: 22 Jan 2018

Author’s response to reviews:

Date 22 Jan 2017

To: BMC Nutrition Editorial

From: Habtamu Temesgen

Subject: A letter accompanying Revision in Response to editor Comments

I am happy to resubmit the revised manuscript “Dietary diversity and associated factors among children aged 6–23months in Sinan Woreda, Northwest Ethiopia: A Cross-Sectional Study” after your fruitful comment and suggestions for publication in BMC nutrition Journal again. Your comment and suggestions are highly significant for the quality of our manuscript and we accept those comments and we tried to address all the comments and suggestions. The corrected manuscript was attached on resubmission without track change. The comment responses were not indicated on line by line rather we correct at the manuscript.

We look forward to receiving your decision.

With regards

Habtamu Temesgen

On the behalf of other authors.